Author’s response to reviews

Title: The mycelium of the Trametes versicolor (Turkey Tail) mushroom and its fermented substrate each show potent and complementary immune activating properties in vitro.

Authors:

Kathleen Benson (kathy@nislabs.com)
Paul Stamets (Paul.s@fungi.com)
Renee Davis (renee.d@fungi.com)
Regan Nally (regan.n@fungi.com)
Alex Taylor (alex.t@fungi.com)
Sonya Slater (sonya.s@fungi.com)
Gitte Jensen (gitte@nislabs.com)

Version: 2 Date: 22 Aug 2019

Author’s response to reviews:

Dear Editor,

Thank you for your feedback and request for two minor clarifications.

1. The study was reviewed and approved by the Sky Lakes Medical Center Institutional Review Board, a local institution that serves the local research community. There was no other ethics committee involved. Let me know if you need any more clarification.

2. I have added a sentence to the “Funding” section, to explain which coauthors are scientists from the funding body, and that their contributions are explained in the section “Authors’ Contributions”.

I have uploaded the clean (no ‘tracked changes’) manuscript.
Please let me know if there is anything else we can do to improve our manuscript or clarify roles and responsibilities. We hope very much to hear whether our manuscript is now acceptable for publication in BMC CAM.

Sincerely,

Gitte.